Author's response to reviews

Title: Chylous ascites associated with chylthorax; a rare sequela of penetrating abdominal trauma: a case report.

Authors:

Joseph M Plummer (joseph_plummer@yahoo.com)
Michael E McFarlane (michael.mcfarlane@uwimona.edu.jm)
Archibald H McDonald (archibald.mcdonald@uwimona.edu.jm)

Version: 8  Date: 7 November 2007

Author's response to reviews:

The Editorial Team
Journal of Medical Case Reports

Dear Sirs,

Re: Manuscript ID 1184023106147998

The abovementioned manuscript is hereby re-submitted for publication and the following formatting changes have been made as listed below.

The words 'Authors' and 'Address' have been removed from the title page.

The author listing has been made identical as they appear in the submission system.

The title of the manuscript was removed from above the abstract.

The Author's contribution sub-section has been altered to reflect the contributions of each author and the patient consent is included under a separate section.

The references have been aligned to the left and all tracking highlights have been removed from this latest version.

Sincerely,

Joseph Plummer.